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  Device Service Record Form for NR-4
	Date
	Comments

(Include the following in each MSR entry: What, Where, Why)
	Info Provided By

	12/07/09
	At ANL for EP, HPR, assembly sequence
	Damon Bice

	3/26/10
	Cavity disassembled at ANL and brought to ICB awaiting transport to ABLE
	Damon Bice

	4/30/10
	Cavity was transported from ICB to ANL per G. Wu’s request
	Damon Bice

	5/10/10
	Cavity received HPR / Assy sequence and was transported to IB#1
	Damon Bice

	6/17/10
	At IB1 still has been tested and vented up ready for pickup
	Morgan Carter

	6/23/2010
	Cavity was transported from IB1 to IB4, ready for disassemble
	Genfa Wu

	7/6/2010
	Cavity disassembled, Al-Mg seal external surface has excessive oxidation sign, from baking?. Cavity transported to ICB, ready for optical inspection
	Genfa Wu

	7/7/2010
	Cavity optically inspected, found a small pit near equator between T-map sensor location 5 and 6.
	Genfa Wu

	8/10/2010
	Cavity transported to ANL for ultrasonic degreasing , cleaning and bagging for pending H Bake out
	Genfa Wu

	10/26/10
	Cavity received U.S. cleaning, dry, bag, and transport to FNAL for H bake out
	Damon Bice

	6/21/11
	Cavity transported to IB4 furnace and received 800c bake w/ 2 hour soak
	Damon Bice

	6/22/11
	Cavity transported from IB4 to ANL
	L. Zimmerman

	8/1/11
	Cavity received 30um EP / HPR / Assy 
	Damon Bice

	8/1/2011
	Cavity transported from ANL to IB1 per Brent Stone @ 1115 hours
	Rick Espinoza

	8/3/11
	Cavity received at IB1
	D. Marks

	8/18/11
	Cavity has been let up, crated, and ready for pick-up
	D. Marks

	8/18/2011
	Cavity has been moved from IB1 to ICB per Dan Marks @ 1545 hours
	RE / LZ

	9/1/11
	A. Dzyuba, pit remelting.  See Single Cell Scheduling Aug 20 2011-CMG for details.
	Jim Rife per Single Cell Scheduling Aug 24 2011-CMG

	
	
	

	
	
	

	10/24/2011
	Cavity transported from ICB to ANL per Damon Bice @ 1015 hours.
	Rick Espinoza

	10/27/11
	Received 20um EP / HPR / Assy sequence and transported to IB1
	Damon Bice

	11/8/2011
	^^^ The previous input is incorrect according to a letter from Allan
	Evgeny Toropov

	11/8/2011
	Moved from IB1 to ICB optical inspection area
	Evgeny Toropov

	11/9/2011
	Optical inspection performed.
	Evgeny Toropov

	11/9/2011
	In the images you can see ~10 cm wide region with traces from silicon replica in the spot of the quench
	Evgeny Toropov

	12/16/2011
	Cavity transported from ICB to ANL.
	N. Bremer

	1/6/2012
	Cavity received 30 µm EP at ANL
	R. Murphy

	01/09/2012
	Cavity transported from ANL to ICB
	N. Bremer

	1/17/11
	Cavity received some laser remelting and was transported to ANL for HPR / VTS assy
	Damon Bice

	1/26/12
	Received standard HPR / Assy and transported to VTS
	Damon Bice

	2/2/12
	VTS test:

Copy from e-log entry by Joe:

TE1NR004 was quench limited to 32 NV/m, also showing strong HFQD beginning at ~28MV/m. Cavity was not baked at 120C - but probably should have been. No FE. Low field Q was 2.4E10. Quench origin detected by FTS.
	Alex Melnychuk

	2/8/12
	Cavity transported from IB1 to ICB
	LZ / JB

	8/22/12
	Cavity location verification.  Cavity @ ICB.
	Jim Rife

	1/14/2013
	Cavity transported from ICB to ANL
	Rick Espinoza

	1/28/2013
	Cavity transported from ANL to IB1 @ 0930 hours
	Rick Espinoza

	2/1/13
	VTS test

Copy from e-log entry by Dmitri:

Cavity te1nr004 appears to have a gross leak. The resonance freq of this cavity is shifted down by ~40 MHz which suggests that cavity filled with liquid  helium.
	Alex Melnychuk

	2/5/13
	Moved to IB4 for cold leak investigation
	Damon Bice

	2/18/2013
	Cavity transported from IB4 to ANL
	Rick Espinoza

	2/26/13
	Cavity received HPR/assembly sequence at ANL
	Chris Baker

	3/1/13
	VTS test:

Copy from e-log entry by Dmitri:

te1nr004 quenches at 22MV/m 

Q looks "normal" for 1.3GHz 1-cell cavity 

(~2e10 at lower E), no X-rays --  

see attached picture.(link to e-log)
	Alex Melnychuk

	3/4/13
	Cavity moved from IB1 to ICB
	JB/LZ

	3/5/13
	Cavity moved to IB4 CPL
	Chad Thompson

	3/7/13
	Electropolished for 2hr 45min, rinsed system with 7.5L Sulfuric acid, average temperature:  22 C EQ/12 C BT, average current:  29.0 A, average flow rate:  0.5 gal/min, estimated material removal:   µm EQ/  µm BT, average modulation amplitude of current oscillations:  60%;  volume of electrolyte used:  26L

Data location:  Q:\TD_SCRF\IB4 Single-Cell CPF\EP\EP Data\2013-03-07 TE1NR004.xlsx
	Chad Thompson

	3/7/13
	Cavity was wiped on the inside with a cleanroom towel soaked in liquinox soap, ultrasonically cleaned for 60min with 0.5% liquinox and UPW, and stored in a bucket of UPW  
	Chad Thompson

	3/7/13
	Cavity received 1 pass HPR
	Chad Thompson

	3/8/13
	Moved cavity to ICB for optical inspection then returned to IB4 CPL tumbling area.
	G.Steuer

	3/12/13
	Moved cavity to IB4 cleanroom for 1 hour US cleaning.  Rinsed with UPW and let hang to dry.
	Chad Thompson

	3/20/13
	Cavity transported from IB4 to ANL
	Chris Baker

	3/20/13
	Cavity received at ANL
	Chris Baker

	3/22/13
	Cavity received HPR/assembly sequence at ANL
	Chris Baker

	3/25/13
	Cavity transported from ANL to IB1
	Chris Baker

	4/8/13
	Cavity transported from IB1 to ICB
	LZ / JB

	4/10/13
	Cavity location verification.  Cavity at ICB.
	Jim Rife

	4/11/13
	Cavity transported to CPL cleanroom for disassembly
	Damon Bice

	1/14/14
	Cavity location verification.  Cavity at IB4.
	Jim Rife

	1/17/14
	Bare Cavity with end caps.
	Rick Espinoza

	2/13/15
	Cavity location verification.  Cavity at IB4
	Jim Rife

	6/22/15
	Location update @ ICB
	Damon Bice
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